Author's response to reviews

Title: How well do second-year students learn physical diagnosis? Observational study of an objective structured clinical examination (OSCE)

Authors:
- Claus Hamann (chamann@partners.org)
- Kevin Volkan (kevin.volkan@csuci.edu)
- Mary B Fishman (MBF3@gunet.georgetown.edu)
- Ronald C Silvestri (rsilvest@caregroup.harvard.edu)
- Steven R Simon (steven_simon@hms.harvard.edu)
- Suzanne W Fletcher (suzanne_fletcher@hms.harvard.edu)

Version: 4 Date: 22 Dec 2001

Dear Dr. Mongin-Bulewski,
Please find attached our manuscript revised as you recommended. Two technical issues remain, from my perspective. First, I cannot crop the Figure in Acrobat Reader despite efforts to resize as directed in AR Help. Second, I cannot remove the extra lines at the bottom of Table 3. I welcome your suggestions as to how I might accomplish these details, if important to the publication of our manuscript. Please advise,
Sincerely,
Claus Hamann